Author's response to reviews

Title: Lymph node enlargement after definitive chemoradiotherapy for clinical stage I esophageal squamous cell carcinoma

Authors:

Yoshito Hayashi (y.hayashi@gh.med.osaka-u.ac.jp)
Tsutomu Nishida (tnishida@gh.med.osaka-u.ac.jp)
Masahiko Tsujii (mt@gh.med.osaka-u.ac.jp)
Shusaku Tsutsui (stsutsui@hosp.itami.hyogo.jp)
Katsumi Yamamoto (katsumiy770@yahoo.co.jp)
Fumiaki Isohashi (isohashi@radonc.med.osaka-u.ac.jp)
Makoto Yamasaki (myamasaki@gesurg.med.osaka-u.ac.jp)
Hiroshi Miyata (hmiyata@gesurg.med.osaka-u.ac.jp)
Motohiko Kato (motohiko@gh.med.osaka-u.ac.jp)
Takuya Yamada (yamtak@gh.med.osaka-u.ac.jp)
Shinichiro Shinzaki (shinzaki@gh.med.osaka-u.ac.jp)
Hideki Iijima (hiijima@gh.med.osaka-u.ac.jp)
Kazuhiko Ogawa (kogawa@radonc.med.osaka-u.ac.jp)
Yuichiro Doki (ydoki@gesurg.med.osaka-u.ac.jp)
Tetsuo Takehara (takehara@gh.med.osaka-u.ac.jp)

Version: 2 Date: 3 September 2013

Author's response to reviews: see over
Dear Prof. Dafne Solera

Thank you for pointing out the necessity to revise tables about confidentiality in our manuscript 9375087910713083 entitled: “Lymph node enlargement after definitive chemoradiotherapy for clinical stage I esophageal squamous cell carcinoma” for consideration of publication in BMC Cancer.

We have modified according to your suggestion. Specifically, we have changed the patient number to simple number in table 2 and 3 in order to be impossible to link patient identification.

Please teach us if further modification is needed.

Thank you for your e-mail and we believe that our work is worthy of publication in BMC Cancer.

Corresponding author: Tetsuo Takehara, MD, PhD
Mailing address: Department of Gastroenterology and Hepatology, Osaka University Graduate School of Medicine, 2-2 Yamadaoka, Suita, Osaka 565-0871, Japan
Phone number: +81-6-6879-3621  Fax number: +81-6-6879-3629
 e-mail address: takehara@gh.med.osaka-u.ac.jp

Sincerely yours,
Tetsuo Takehara